Author’s response to reviews

Title: TCF7L2 and therapeutic response to sulfonylureas in patients with type 2 diabetes

Authors:

Andreas Holstein (Andreas.Holstein@t-online.de)
Michael Hahn (mhahn78@googlemail.com)
Antje Körner (antje.koerner@medizin.uni-leipzig.de)
Michael Stumvoll (michael.stumvoll@medizin.uni-leipzig.de)
Peter Kovacs (peter.kovacs@medizin.uni-leipzig.de)

Version: 4 Date: 15 February 2011

Author’s response to reviews: see over
Dear BMC Series Editorial Production Team,

according to your suggestions, we now made following major and minor revisions:

1) revised Author's contributions based on the provided example,

2) corrected citation for reference 17,

3) renamed Methods,

4) removed track changes,

5) removed visible vertical lines from our tables.

Please, let us know if there is anything else we could help with.

Many thanks and best wishes,

Peter Kovacs